Author's response to reviews

Title: Bereavement care interventions: a systematic review

Authors:

Amanda L Forte (forte@email.chop.edu)
Malinda Hill (hillme@email.chop.edu)
Rachel Pazder (pazder@rosemont.edu)
Chris Feudtner (feudtner@email.chop.edu)

Version: 4  Date: 20 July 2004

Author’s response to reviews:

20 July 2004

Dear BMC Editorial Team:

Thank you for message regarding "Bereavement care interventions: a systematic review."

In response to your request for formatting changes, we made all the suggested versions with one exception: we did not provide web-site addresses for the databases (MEDLINE, etc) that we consulted, since most of these databases are proprietary and one gains access to them (typically) through a variety of library portals. Checking other systematic reviews published by BMC journals, I do not find them attempting to cite the databases, and while I welcome new ways to help readers, in this instance I think it would be confusing (the best advice being: go talk to a reference librarian).

I also note that all common terminology that we'd want for PubMed indexing is indeed in the title and abstract.

Please let me know if you have any further questions or issues regarding the manuscript.

We are pleased that the study will appear under your auspices, and grateful for the time you have taken to improve the quality of the report.

Sincerely,

Chris Feudtner, MD PhD MPH
Director of Research and Attending Physician
Pediatric Advanced Care Team at CHOP
Tel: 267.426.5032 Email: feudtner@email.chop.edu